Author’s response to reviews

Title: Muscle recruitment patterns during the prone leg extension test.

Authors:

Gregory J Lehman (glehman@cmcc.ca)
Duane Lennon (duane_lennon@hotmail.com)
Ben Rayfield (rayfield@hotmail.com)
Mike Poschar (poshdc@hotmail.com)
Brian Tressider (bjt75@hotmail.com)

Version: 3 Date: 6 Feb 2004

February 6, 2004

Dear BMC Editors,

I have made all the necessary editorial changes.

Reviewer #1 had three concerns:

1. ...typos...

All typos were corrected.

2. ...statement regarding within subject variability in results since a point is made in the discussion.

The statement was removed from the discussion as this is extraneous to the major conclusion and intention of the paper.

3. ...a reference as to why 10% of peak muscle activity.

The reference for this value was cited in the methods. This value was used because it was used in previous studies.

Thank you,

Gregory Lehman